Author's response to reviews

Title: Specialist and Primary Care Physicians' Views on Barriers to Adequate Preparation of Patients for Renal Replacement Therapy: A Qualitative Study

Authors:

Raquel C Greer (rfcharle@jhmi.edu)
Jessica M Ameling (jamelin1@jhmi.edu)
Kerri L Cavanaugh (kerri.cavanaugh@Vanderbilt.Edu)
Bernard G Jaar (bjaar@aol.com)
Vanessa Grubbs (grubbsv@medsfgh.ucsf.edu)
Carrie E Andrews (carrie.e.andrews@gmail.com)
Patti Ephraim (pephraim@jhu.edu)
Neil R Powe (NPowe@medsfgh.ucsf.edu)
Julia B Lewis (julial.B.lewis@Vanderbilt.Edu)
Ebele M Umeukeje (ebele.m.umeukeje@Vanderbilt.Edu)
Luis Gimenez (lgimene1@jhmi.edu)
Sam James (JamesS@medsfgh.ucsf.edu)
Ebony Boulware (ebony.boulware@duke.edu)

Version: 4
Date: 26 January 2015

Author’s response to reviews:

Dear Editors,

Thank you for accepting our manuscript, entitled “Specialist and Primary Care Physicians' Views on Barriers to Adequate Preparation of Patients for Renal Replacement Therapy: A Qualitative Study” (MS: 1824957840147603), for publication in your journal. We have made the requested revisions, and the point-by-point response is provided below:

Sincerely,

Raquel C. Greer, MD, MHS
Corresponding author

Response to Editors’ Requests:

1. Need to ensure that all the authors are mentioned in the Authors' Contributions (currently only 11 are present).

All 11 authors are now mentioned in the Authors’ Contributions section.

2. Second copy of the manuscript needs to be removed from the additional files

The second copy of the manuscript has been removed.